Author's response to reviews

Title: Maternal lifestyle characteristics (physical activity, smoking habits and alcohol consumption) during pregnancy, and the risk of obesity in the offspring: a study of 5,125 children

Authors:

Stamatis Mourtakos (stmourt@gmail.com)
Konstantinos D. Tambalis (dp425603@hua.gr)
Demosthenes B. Panagiotakos (dbpanag@hua.gr)
George Antonogeorgos (gantonogeorgos@gmail.com)
Giannis Arnaoutis (giannis_arnaou@hotmail.com)
Labros S. Sidossis (lasidoss@utmb.edu)

Version: 2
Date: 27 November 2014

Author's response to reviews: see over
Dear Sir/ Madame,

Further to your previous email I have introduced the following changes to the manuscript:

1. Line Numbering and page numbers
2. Competing interests
3. Authors’ contribution
4. Change of font style to Calibri
5. Title Page to include email addresses of all authors.

Please inform me if you require anything else before it can be sent for peer review. Thank you very much for your attention.

Yours sincerely,

Stamatis Mourtakos